Author’s response to reviews

Title: Impact of early applied upper limb stimulation: The EXPLICIT-stroke programme design.

Authors:

Gert Kwakkel (g.kwakkel@vumc.nl)
Carel G.M. Meskers (c.g.m.meskers@lumc.nl)
Erwin E. van Wegen (e.vanwegen@vumc.nl)
Guus Lankhorst (gj.lankhorst@vumc.nl)
Alexander C.H. Geurts (a.geurts@reval.umcn.nl)
Annet A. van Kuijk (a.v.kuijk@tolbrug.nl)
Eline Lindeman (E.lindeman@pmbr.azu.nl)
Anne Visser-Meily (j.m.a.visser-meily@pmbr.azu.nl)
Erwin de Vlugt (e.devlugt@tudelft.nl)
Hans J.H. Arendzen (j.h.arendzen@lumc.nl)

Version: 2 Date: 26 October 2008

Author’s response to reviews:

Editor
BMC Neurology

Dear Editor,

Regarding our paper, the protocol of the EXplaining PLastICITy after stroke study: “Impact of early applied upper limb stimulation: The EXPLICIT-stroke programme design”, we have now included a statement on the Medical Ethical approval of our study in the "Methods" section as you requested.

We trust that this change in the manuscript will be of satisfaction to you and that our manuscript can now be processed further.

Kindly awaiting your decision,

sincerely yours,

Prof dr. G. Kwakkel
Dr. C.G.M. Meskers
on behalf of the EXPLICIT-stroke trialists
www.explicit-stroke.nl